University of Toronto, Department of Electrical and Computer Engineering

ECE241F - Digital Systems TAs Who Marked Each Midterm Question

Fall 1999 A. Carusone, P. Lehn, J. Rose If you have a query concerning your mark on the midterm, contact one of the TAs below who marked the question. They will consult the photocopies of the midterms that we have made (not the original that you bring).To find out which room the TA is in on the lab day, see the TA link on the main course web page.

Questions Mark ed TA Email Lab Day 1 Farhang Vessal fvessal@vrg Monday &Thurs 1 Tooraj Esmailian tooraj@eecg Monday &Thurs 1 Yucai Zhang zyc@vrg Thursday 2, 3(b) Mike Sheng sheng@eecg Monday 2, 3(b) Suneil Sastry sastri@ecf Monday 2, 3(b) Ryan Wu wussu@ecf Thursday 3(a) Daniel Lopez dlopez@comm Monday &Thurs 3(a) Farhang Vessal fvessal@vrg Monday &Thurs 3(a) Tooraj Esmailian tooraj@eecg Monday &Thurs 4 Desh Singh singhd@eecg Monday 4 Edward Yu edyu@vrg. Thursday 4 Kostas Pagiamtzis pagiamt@eecg Thursday 5 Andrew Eckford eckford@comm Monday 5 Prabhakaran Selvaduri sprabha@comm Thursday 5 Tim Demonte demonte@ibme.utoronto.ca Monday &Thurs